Author's response to reviews

Title: Problems and needs for improving primary care of osteoarthritis patients: the views of patients, general practitioners and practice nurses

Authors:

Thomas TR Rosemann (thomas_rosemann@med.uni-heidelberg.de)
Michel MW Wensing (michel_wensing@med.uni-heidelberg.de)
Katharina KJ Joest (katharina_joest@med.uni-heidelberg.de)
Matthias MB Backenstrass (matthias_backenstrass@med.uni-heidelberg.de)
Cornelia CM Mahler (cornelia_mahler@med.uni-heidelberg.de)
Szecsenyi JS Joachim (joachim_szecsenyi@med.uni-heidelberg.de)

Version: 4  Date: 18 May 2006

Author's response to reviews:

As required we added information about the ethical approval at the end of the methods section, including the approval number. We also revised the conflict of interests statement and changed the formatting of the tables 2-4.